DNB Working Paper

No. 554 / April 2017

Pension funds' carbon footprint and investment trade-offs

Martijn Boermans and Rients Galema

DeNederlandscheBank

EUROSYSTEEM

Pension funds' carbon footprint and investment trade-offs

Martijn Boermans and Rients Galema *

* Views expressed are those of the authors and do not necessarily reflect official positions of De Nederlandsche Bank.

Working Paper No. 554

De Nederlandsche Bank NV P.O. Box 98 1000 AB AMSTERDAM The Netherlands

April 2017

Pension funds' carbon footprint and investment trade-offs*

Martijn Boermans^{a,b} and Rients Galema^b

^a De Nederlandsche Bank, Amsterdam, The Netherlands ^b Utrecht University School of Economics, Utrecht, The Netherlands

10 April 2017

Abstract

With the adoption of the Paris Agreement in December 2015, better understanding of portfolio carbon dioxide (CO_2) exposures has become increasingly important for investors, regulators and society at large. In this paper we measure the portfolio carbon footprints (CFP_s) of pension funds' stock investments. We utilize security-by-security holdings of Dutch pension funds over the period 2009-2015 and combine this with firm-level CO_2 information to analyze the drivers of CFP at the portfolio level. The results show that pension funds face intricate trade-offs when aiming to reduce portfolio related CO_2 emissions: expected dividend yields are positively related to the carbon footprint. The dividend trade-off with carbon exposures only applies to well-funded pension funds and is driven to some extent by investments in energy and utility companies. For the period 2013-2015 pension funds which publicly disclose their carbon footprint tend to have relatively lower exposure to firm with high carbon emissions.

Keywords: carbon emissions, pension funds, institutional investors, socially responsible investment, climate change.

JEL classifications: G11, G23, H55, Q54, Q56.

^{*} Without implicating we thank Bart Bos, Raymond Chaudron, Lammert-Jan Dam, Patty Duijm, Jakob de Haan, Robert Vermeulen, Utz Weitzel and participants at the De Nederlandsche Bank research seminar (2016) and Utrecht University School of Economics internal seminar (2017) for comments and suggestions on an earlier version of this manuscript. All remaining errors are ours. Views expressed are those of the authors and do not necessarily reflect official positions of De Nederlandsche Bank.

Corresponding author: m.a.boermans@dnb.nl.

1 Introduction

In December 2015 world leaders signed the Paris Agreement on Climate Change and thereby committed themselves to significantly reducing carbon dioxide (CO_2) emissions. To achieve these goals, firms must reduce their carbon footprint (CFP). Similarly, many institutional investors have started to measure and publicly disclose the CFP of their investment portfolios with the aim of lowering the carbon emissions associated with their investments.¹

Institutional investors are interested in their CFP for two main reasons. First, they act as "universal owners" because of long-term and diversified investment positions in a wide range of firms (Dimson et al., 2015). These assets under management face a trade-off between expected short-term gains from carbon-intensive businesses versus potential long-term benefits from a decarbonized economy. That is, institutional investors recognize the importance of fossil fuels for the near future, but are also confronted with environmental and regulatory risks associated with carbon-intensive investments, such as the risk of stranded assets and long-term tail risks associated with catastrophic events related to climate change (Andersson et al., 2016; Griffin et al., 2015; Hong et al., 2016).

In addition, institutional investors face pressure from stakeholders to reduce their investments in carbon-intensive stocks. Pension funds in particular are likely to face a high degree of public scrutiny regarding the social responsibility of their investments. This resulted in strong engagement by pension funds in socially responsible investment as they have been a powerful force in shaping company behavior (Dyck et al., 2016; McCahery et al., forthcoming; Sievanen et al., 2013). More generally, the market for sustainable investment has grown significantly in the past decade (Bohl et al., 2015).

It is a priori unclear whether investors like pension funds necessarily face a trade-off

¹In recent years, numerous investors have joined a wide range of initiatives that aim to reduce investors' *CFP* including the Carbon Tracker Initiative, the Climate Disclosure Standard Board (CDSB), the Montréal Carbon Pledge, the Portfolio Decarbonization Coalition (PDC) and the United Nations Principles for Responsible Investment (PRI).

between low-carbon investments and returns. In general, literature on socially responsible investment (SRI) has yielded mixed results regarding the trade-offs between returns and social responsibility (e.g. Galema et al., 2008; Renneboog et al., 2008) as the sign and significance of the relationship depends strongly on the specific dimension of social responsibility (Ziegler et al., 2007). The same holds for the environmental dimension. Restricting the investment universe to stocks of firms with low CO_2 emissions hampers diversification and decreases expected risk-adjusted returns (Bauer et al., 2005; Kreander et al., 2005; Scholtens, 2006). This effect is abated if CO_2 emissions are highly concentrated in very few firms (Koch and Bassen, 2013). Indeed, investors can construct stock portfolios with lower CFP while following a market benchmark with negligible tracking error (Andersson et al., 2016).²

In this paper we focus on the environmental dimension of SRI—specifically the CO_2 exposures related to pension funds' stock investments. To investigate why some pension funds have a higher CFP than others we study Dutch pension funds, which have about 1.3 trillion euro under management. Proprietary security-by-security data from the Dutch central bank (DNB) on equity investment are used for the period 2009Q1-2015Q4. These data are matched with CO_2 emissions from the Thomson ASSET4 database and supervisory data on pension funds' equity investment. That is, like Andersson et al. (2016) we aggregate firm-level CO_2 emissions from security-by-security (stock) holdings data to the pension funds' portfolio level.

Our main results highlight the significant investment trade-offs faced by pension funds when reducing portfolio CO_2 exposures. First, there is a trade-off between the portfolio's carbon dioxide emissions and the expected dividend returns. This raises a concern because pension funds typically have a strong preference for high-dividend stocks (Allen et al., 2000; Short et al., 2002; Dahlquist et al., 2014; Bams et al., 2016). We

 $^{^{2}}$ De Jong and Nguyen (2016) show that something similar holds for bond indexes.

find that the trade-off is to some extent driven by investments in firms from energy and utilities industries, which is in line with Andersson et al. (2016). Moreover, we only observe this trade-off for well-funded pension funds, suggesting they reduce CFPwhen they deem it affordable, which is consistent with other studies on pension funds' investment behavior (An et al., 2013) and the role of regulatory requirements (Andonov et al., 2015). One interpretation of this result is that reducing CFP is considered a luxury. Second, our main results show that there is a negative relationship between pension funds' active share (Cremers and Petajisto, 2009)—a measure that indicates the extent to which pension funds actively managing their portfolio by deviating from the benchmark—and their CFP. This suggests that pension funds reduce CFP by deviating from the market portfolio, i.e. lowering CFP is associated with higher market risk. Third, public pressure could be a possible motivation for reporting and subsequently reducing CFP. Indeed, we find that pension funds that reported CO_2 portfolio exposures had much lower CFPs.

This paper makes three contributions. First, we contribute to a small but growing literature on institutional investors and corporate social responsibility (CSR). This literature has focused mainly on how firms' CSR is affected by institutional holdings (e.g. Dyck et al., 2016; Dimson et al., 2015), whereas we investigate how institutional investors' CSR is affected by the portfolio of firms they invest in. Second, a related literature analyzes the consequences of reducing CFP for a hypothetical portfolio (e.g. Andersson et al., 2016; De Jong and Nguyen, 2016). By contrast, we use actual pension fund holdings to calculate portfolio level CFP. Third, we contribute to the literature on green investments, which mainly analyzes the performance of green compared to conventional mutual funds.³ Instead of classifying funds dichotomously, our approach allows us to directly measure and explain carbon-related environmental performance.

³See e.g. Cortez et al. (2009, 2012); Climent and Soriano (2011); Nofsinger and Varma (2014); Ibikunle and Steffen (forthcoming); Rezec and Scholtens (2016); Silva and Cortez (2016); Shin et al. (forthcoming) and (e.g. Derwall et al., 2005; Ziegler et al., 2007; De Haan et al., 2012) who use replicating portfolios techniques that vary along the environmental dimension.

This paper is organized as follows: Section 2 presents the security-by-security holdings data, the firm-level data and the pension fund data. Section 3 explains how carbon exposures are measured and introduces our estimation model. Section 4 provides the main results. Section 5 presents some robustness tests for our carbon exposure measure. Section 6 concludes.

2 Data

We combine several data sources to analyze the carbon footprints (CFP) of pension funds. First, the Thomson Reuters ASSET4 database is used to obtain quarterly CO_2 emission levels, CE_{jt} for 5,260 firms for 2009-2015. The CO_2 emissions data are taken from company reports in line with the Greenhouse Gas Protocol, Kyoto Protocol or EU Trading Scheme where the reported Greenhouse Gas Protocol score is the preferred measure.⁴ If a firm does not report carbon emissions, Thomson Reuters estimates them using their Carbon Data & Estimate Model (Thomson Reuters, 2016). These data are linked to firm-level characteristics from Datastream. Finally, at the firm-level, industry NACE classifications are obtained from the ECB Centralised Securities Database (CSDB).

Second, information on pension fund investments is obtained from a confidential dataset from the Dutch Central Bank (DNB). Data from the balance of payments and securities statistics department contain quarterly security-by-security portfolio investment positions in stocks of pension funds. In addition to pension fund holdings, we incorporate mutual funds data because a small number of pension funds hold substantial amounts of fiduciary investment through mutual funds. We use security-by-security stock holdings of 16 mutual funds so that we can 'look through' the pension fund share-holdings in these mutual funds and increase the coverage of the stock positions. Based

⁴The CO_2 emissions include direct and indirect equivalents of emissions, i.e. Scope 1 and Scope 2.

on the participation shares, we proportionally attribute the mutual fund positions for each stock to the pension fund's portfolio.⁵ After merging the firm-level data with the pension fund holdings we observe the stock holdings of 41 pension funds for 4,657 unique stocks over the period 2009Q1-2015Q4. These security-by-security stock holdings consist of 1,476,385 observations for each pension fund's stock position over time.

The granular data is collapsed to the pension fund portfolio level. Pension fund characteristics are taken from aggregate supervisory reports to DNB.⁶ Overall, the coverage of the security-by-security data on portfolio investments is high compared to the aggregated reported stock positions in the pension funds balance sheet items from the supervisory reports (on average 82.4 percent, see A.1). The coverage generally improves over time but is very constant across smaller and larger pension funds.⁷ Finally, manually collected information from pension funds' websites and annual reports are used to observe which pension funds report their carbon emissions.

3 Method

3.1 Measuring carbon footprint of investments

To measure the carbon footprints of the stock investments for each pension fund a relative carbon efficiency measure of the portfolio is obtained, CFP_{it} . This variable is based on the CO_2 emissions of a firm j adjusted for the firm's output and the relative importance of the stock position in firm j in the overall portfolio of pension fund i at time t. Firm j's carbon (in)efficiency CI_j is derived as a ratio of CO_2 exposure in tonnes to sales in million euros in line with the literature and practice (Andersson et al., 2016; PGGM,

⁵See also Broeders et al. (2016) who use similar data to study pension fund herding behavior or more generally Ferson and Khang (2002) who construct performance measures of pension funds using portfolio holdings data and Bradley et al. (2016) how study firm-level characteristics of pension funds' portfolios to study political influences.

⁶See also de Dreu and Bikker (2012), Duijm and Steins Bisschop (2015) and De Haan (forthcoming) who use similarly aggregated data to study pension fund investments in general.

 $^{^{7}}$ By the end of 2015, total stock holdings of the pension funds in our final sample were 386 billion euro.

2016). CFP_{it} is calculated as the portfolio weighted average of the firms' CO_2 efficiency ratios.

Specifically, consider a pension fund i that holds stocks in firm j with an investment position I_{ij} where the firm's carbon inefficiency is CI_j . The CFP_i of each pension fund i is measured as the portfolio weighted average of CI_j in equity investments.

$$CFP_i = \sum_j w_{ij} \times CI_j \tag{1}$$

where $w_{ij} = I_{ij}/(\sum_{j=1}^{n} I_{ij})$ indicates the portfolio weight: the ratio of the market value of each stock investment (I_{ij}) to the total market value of the pension fund's equity investments $(\sum_{j=1}^{n} I_{ij})$. To compute CFP_{it} we take CI_{jt} with quarterly adjusted investment weights w_{ijt} . Thus, CFP_{it} captures the weighted average of how many units of CO_2 are emitted per unit of sales for each euro invested by pension fund i at time t.

Using this method, we can consistently create CFP_{it} across investors' stock portfolios over time.⁸ A key advantage of this measure is that it is a relative carbon efficiency measure and is thus neither affected by the size of the stock holdings of the investor $(\sum_{i=1}^{n} I_{ijt})$ nor by the coverage of stocks in our dataset.⁹

[Insert Table 1—"Summary statistics of carbon efficiency by industry"]

Before applying this method to pension fund portfolios, let us elucidate how CFP varies across industries by inspecting the security-level data from Thomson Reuters. Table 1 presents summary statistics of security-level CI_{ijt} for a panel dataset of all stocks in our sample (2009Q1-2015Q4), which is weighted by the stock's market value

⁸The CFP_{it} scores are very similar to scores of pension funds that reported their CFP in 2014, which validates our measure.

⁹In general the coverage is better for larger firms. A potential bias in CFP_{it} would be created if the (smaller) excluded firms' CO_2 emissions are significantly different from the observed CO_2 emissions in terms of relative output efficiency while at the same time there are strong differences in stock coverage in the cross-section of pension funds' portfolios; note that the coverage is high across all pension funds, see A.1. In addition, these two variations (carbon efficiency and stock coverage) must be correlated with pension fund characteristics to create a bias in our results.

against total industry market value to create a stock-level CFP measure. It shows that the highest CFP firms can be found in the industry 'Electricity, gas and steam', followed by 'Mining and quarrying'. These high CFPs are due to high average firm CO_2 emissions and relatively modest firm sales within these industries.¹⁰ There is considerable withinindustry variation in CFP, although also the extent of this variation differs considerably across the industries.

To better understand the general market context of our CFP measure for the Dutch pension funds, we apply Equation (1) to calculate the CFP of a benchmark portfolio, $CFP^{Benchmark}$. The benchmark consists of all stocks in the ASSET4 universe and captures the carbon (in)efficiency of a market portfolio with an average underlying stock market value of 31 trillion euro. Figure 1 compares the average market $CFP_t^{Benchmark}$ to the average CFP_{it} of the pension funds in our sample. Clearly the average pension fund's CFP is much lower than that of the general market. Dutch pension funds perform better than the benchmark across the whole sample period and this differential is constant over time.¹¹ In addition, the $CFP^{Benchmark}$ has become much less carbon-intensive over time. In 2009Q1 the average $CFP^{Benchmark}$ was 440 and this decreased to 250 in 2015Q4: a 43% reduction.¹²

[Insert Figure 1—"Average pension fund footprint versus a market benchmark"]

The shaded area in Figure 1 shows the 10^{th} to 90^{th} percentiles of the CFP distri-

¹⁰Appendix B provides summary statistics for the different dimensions of our security-level data. To illustrate, Table B.1 shows that on average the largest firms in our sample in terms of market values are in 'Manufacturing'. Table B.2 indicates that pension funds also have the highest stock investments in 'Manufacturing', whose firms also have highest average output in terms of sales (see B.3). By contrast, B.4 shows that the highest absolute CO_2 emissions (not scaled by sales) are due to 'Electricity, gas and steam'.

¹¹Unreported t-tests shows that this difference is significant at 1%. In addition, a panel unit root test shows that CFP is (trend) stationary.

¹²Indeed over time firms on average improved their carbon efficiency, however, the large reduction also stems from a compositional shift in the global market portfolio: High-tech firms with relatively low CO_2 emissions rapidly increased their market values (and thus their weights in the benchmark), while firms with high CO_2 emissions have witnessed relatively poor stock performance (Ibikunle and Steffen, forthcoming).

bution, which indicates significant CFP variation across pension funds at each point in time. Our estimation—to which we will turn to next—attempts to capture this variation while controlling for the time trend by including year-quarter fixed effects.

3.2 Estimation method

We estimate the following equation to explain the carbon footprint (CFP) of pension fund's *i* stock portfolio at time *t*:

$$CFP_{it} = \beta_0 + \beta_1 X_{it-1} + \beta_2 Year\text{-}quarter_t + \epsilon_{it}$$

$$\tag{2}$$

where X_{it-1} indicates our explanatory variables. Because there is a clear time trend in the $CFP^{Benchmark}$ we include year-quarter fixed effects to control for unobserved heterogeneity of CFP_{it} over time. Standard errors are clustered at the pension fund level.

Table 2 provides the summary statistics of the variables used in this study. The average absolute CO_2 emission level footprint is 1.6 million tonnes as measured by the weights of the pension funds' investments in the firm. Our dependent variables CFP_{it} averages 288.3 tonnes CO_2 per million euro of sales (Std.=76.4).

Our main estimations include the following variables, which are all lagged by one quarter to prevent simultaneity by construction. To capture pension fund size, log $TotalAssets_{it}$, we obtain total assets from the aggregate supervisory balance sheet items. We include several stock characteristics: previous four quarter average stock's $Returns_{it}$, $DividendYield_{it}$ and β_{it} .¹³ We split the total stock returns in percentage price changes over the previous year, $Return_{it}$, and $DividendYield_{it}$. We winsorize $Return_{it}$, $DividendYield_{it}$ and β_{it} at 1% and aggregate all stock characteristics to the portfolio level using the market value of the stocks to weight the total stock portfolio.

¹³Note that Thomson Datastream computes the β based on monthly price movements of the share over a five year period compared to the Datastream total market index for the country.

Of the pension funds' portfolios, the yearly average stock returns are 4.3% plus a dividend yield of 2.9%. With an average number of stocks of around a thousand, pension funds hold broad stock portfolios, which is also evident from the average market β of 1 (Std. = 0.1).

One concern is that pension funds simply follow a market benchmark, accepting whatever CFP is implied by the benchmark. Cremers and Petajisto (2009) introduce *ActiveShare*, which measures the extent to which investors deviate from the investment benchmark. It consist of summing the absolute values of differences of investment weights with the benchmark and dividing by two, where the benchmark is the same as defined above. An *ActiveShare* of zero implies completely following the benchmark, while an Active share of 100% implies completely deviating from it. This measure is closely related to the tracking error (Andersson et al., 2016). We include *ActiveShare_{it}* to test whether pension funds shift their portfolio to increase or reduce *CFP*. Table 2 shows that the mean of *ActiveShare_{it}* is 35%, which implies that on average pension funds have a tendency to track the benchmark, but not very strongly.

Next, we obtain several pension fund level characteristics using aggregated quarterly supervisory data. First, we calculate the share of equity $ShareEquity_{it}$ by taking total stock holdings from the balance sheet as a percentage of total assets. One average, 25% of the assets are allocated to stock investments, with considerable variation across pension funds and over time. These stock level and pension fund level characteristics will be used in our main regressions.

Pension funds report their funding ratios, $FundRatio_{it}$, which is a measure of the assets the pension fund holds against its long-term liabilities.¹⁴ In the Netherlands there are regulatory requirements that aim to safeguard the financial health of pension funds mainly through avoiding being underfunded, where the required minimum funding

¹⁴The funding ratio is defined as the market value of the assets over the market value of the liabilities, which is calculated as the present value of future pension benefits using the term structure of the risk-free market interest rate (like a swap rate) as discount factor De Haan (forthcoming).

ratio is 105% (de Dreu and Bikker, 2012; Andonov et al., 2015; De Haan, forthcoming). This limit is set by the supervisor and once it is breached, pension funds must ensure they take precautionary actions against underfunding as set out in their mandatory recovery scheme. With an average funding ratio of 110%, pension funds in the sample are sufficiently funded. The indicator variable $FundRatio_{it} > 1.05$ equals one if the funding ratio exceeds 105% and is zero otherwise, indicating that there is a funding deficit (which occurs in 40% of our observations). The indicator variable $\Delta_+FundRatio_{it} > 1.05$ equals one once an initially underfunded pension fund reaches a funding ratio that exceeds minimum regulatory funding ratio of 105%, and zero otherwise. On average over time 10% of the pension funds experience such a situation. Funding ratio data will be used in an extended specification model.

 $CarbonReporting_{it}$ is defined as an indicator variable equal to one when pension funds report carbon emissions in their annual report or on their website and zero otherwise. Table 2 shows that 19% of the observations in our unbalanced panel data report CFP. Note that the reporting only started recently and will therefore be analyzed separately.¹⁵

Finally in a distinct analysis we control for contemporaneous industry investment weights (based on market values). Table 2 shows that pension funds hold on average about a third of their equity in industry C ('Manufacturing') and about a fifth in industry K ('Financial and insurance activities').

[Insert Table 2—"Summary statistics"]

¹⁵Appendix D presents a correlation matrix which shows that the correlation coefficients among the explanatory variables are relatively low (i.e. below 0.35) with the exception of — as expected — the association between size and reporting (r = 0.53) and returns and dividends (r = -0.60). Note also that multicollinearity appears no concern as all VIF scores for the regressions run in this paper are well below 10.

4 Results

4.1 Main determinants of pension funds' carbon footprints

Table 3 presents our main estimates. Somewhat surprisingly it shows that larger pension funds have a higher CFP. Interpreting the effect size in terms of a semi-elasticity, a doubling in size (i.e. a 100% increase) is associated with an increase in CFP ranging from 12.79 to 14.67, which is modest compared to an average CFP of 288.3. These results are in line with Hoepner and Schopohl (2016) who find a negative relationship between pension fund size and the probability of signing the Principles for Responsible Investment (PRI). It is inconsistent with the notion that larger pension funds are more concerned about corporate responsibility (Scholtens, 2006) and the finding that larger investors are better able to screen environmental friendly firms due to the monitoring cost involved with managing CFP (Sievanen et al., 2013; Kempf and Osthoff, 2008).

[Insert Table 3—"Main results: Determinants of pension funds' carbon footprints"]

We find no relationship between previous year's portfolio stock returns and CFPs. This finding is consistent with literature that shows that portfolios of firms with better social, governance and environmental scores have risk-adjusted returns similar to that of the market (Margolis and Walsh, 2003; Orlitzky et al., 2003; Bauer et al., 2005; Cortez et al., 2009), although our returns exclude dividend payments.

The average expected dividend yield of the portfolios is strongly related to CFP. The outcomes suggest that pension funds that hold stocks with higher dividend yields tend to have a higher CFP, ceteris paribus. The estimated coefficients imply that a dividend yield rise in the portfolio by one percentage point is associated with a CFP increase of about 30 to 40 tonnes per million euros of sales, which is sizable compared to an average CFP of 289. These results contribute to the debate on the relationship between returns and environmental performance of stock portfolios as some studies suggest a positive

link between returns and social responsible investments (Derwall et al., 2005; Shin et al., forthcoming), while others, in line with our dividend effect, suggest the contrary (e.g. Cortez et al., 2012; Climent and Soriano, 2011; Nofsinger and Varma, 2014; Rezec and Scholtens, 2016; Silva and Cortez, 2016).

Table 3 shows that pension funds with higher portfolio β have a lower *CFP*, ceteris paribus. As a pension fund moves its portfolio β up one standard deviation (0.1), *CFP* is reduced by about 20 tonnes per million of sales. One possible explanation for this relationship is that carbon-intensive industries like energy and utilities are generally less cyclical (Griffin et al., 2015). Also, Ibikunle and Steffen (forthcoming) note that more environmentally friendly mutual funds tend to have greater exposure to smaller firms and growth stocks, which have higher β 's and lower carbon emissions.

In addition to introducing more market risk, pension funds can decide to let their portfolio deviate from the market portfolio as captured by $ActiveShare_{it}$. Here we find a strongly negative relationship between $ActiveShare_{it}$ and CFPs. This suggests that pension funds that deviate from the market portfolio, shift their portfolio to firms that emit less carbon per unit of sales. The estimated effect size is rather strong as a one standard deviation increase in ActiveShare is associated with a decrease in CFP of about 12. Finally, we find no relationship between the share of assets allocated to equity and CFP. This suggests that pension funds' strategic asset allocation does not influence CFP.

Taken together, the combined results suggest that pension funds face stark tradeoffs. Those that have managed to lower their carbon footprint have had to accept lower expected dividend yields and higher expected systematic risk (β). If anything, it seems that larger pension funds are not better positioned to reduce CFPs. Since pension funds seem to lower their CFP by deviating from the market portfolio, larger pension funds might find it more difficult to reduce CFP. Their sheer size may make it more difficult to deviate substantially.

4.2 Carbon footprints, pension funds' funding ratios and dividends

Pension funds must pay very close attention to their funding ratios in order to determine whether they are able to meet future liabilities (Andonov et al., 2015; De Haan, forthcoming). In the Netherlands, the supervisor has regulatory powers to demand an investment recovery scheme of pension funds once they are underfunded, i.e. having a funding ratio of less than the minimum 105%. During the recovery phase pension funds have less leeway to increase their risk profile. Although this applies mostly to their strategic asset allocation, they might also be intrinsically motivated to pursue more conservative investment strategies. For instance, there could be a correlation between financially weaker pension funds and the degree to which pension funds tilt their portfolio towards investments with lower CFP. In the short run, underfunded pension funds might care more about financial returns than CFP reduction. In the main results we find a strong trade-off between pension funds' expected portfolio dividend yield and CFP. In this section we investigate to what extent this trade-off applies when a pension fund's financial situation changes in terms of its funding ratio.

Table 4 augments the main results from Table 3 by further inclusion of various measures of the funding ratio and interactions with the portfolio's overall dividend yields. Column (1) shows that there is no direct relationship between the level of the funding ratio and CFPs. Column (2) includes an interaction term with dividends and these results also do not present any associations with CFP. In Column (3) we include an indicator variable equal to one if a pension fund meets the regulatory minimum funding ratio of 105% and zero otherwise. Again there appears to be no direct relationship between the funding ratio and CFP. These findings seem to show that there is no relationship between funding ratio and CFP.

To study the relation between funding ratio, dividends and CFP further, Column (4) includes the interaction between the indicator for being well-funded with dividend

yield. This association is weakly significant.¹⁶ One interpretation of this finding is that pension funds consider lowering their CFP as a luxury. Once pension funds reach a sufficient level of funding, they could reduce CFP by reconsidering their preferences for dividends. To test explanation further, we include an indicator variable that equals one when a pension fund reaches a funding ratio higher than 105% and zero otherwise in Column (5), that is, as it move in or out being underfunded. In Column (6) we again interact this variable with dividends. The significant and positive interaction between the portfolio's dividend yield and the move towards being over-funded suggests that when pension funds are able to increase their funding ratios above the regulatory minimum, they have room to consider the trade-off between dividend yield and CFP. The marginal effect of $\Delta_+FundRatio_{it} > 1.05$ starts to become positive around the average dividend yield of 2.9%. If pension funds that have recently been able to increase their funding ratio want to decrease CFP, they are able to do so only if they are willing not to pick stocks associated with higher expected dividend yields.¹⁷

The finding that pension funds that are able to build up sufficient financial buffers and aim to reduce CFP face a trade-off between expected stable returns from dividends and reducing their portfolios' CO_2 emission levels is new in the literature. One interpretation of our results is that well-funded pension funds are willing to increase their systematic risk as captured by (β) to reduce CFP. This fits well with Rauh (2009) who shows that poorly-funded pension funds allocate a greater share of pension fund assets to safe assets (i.e. government bonds), while well-funded pension funds are more willing to take risk and invest more heavily in equity. In addition, underfunded pension funds may be unable to sacrifice dividend income (Mohan and Zhang, 2014), which explains the significant interaction effect between the funding ratio and dividend on CFP.

¹⁶Unreported results for a negative delta of the funding ratio do not yield any significant associations.

¹⁷Note that this interaction effect is absent for pension funds which funding ratio moves below 105 percent, which strengthens this explanation. These findings are complementary to our main results concerning the dividend ratios.

[Insert Table 4—"Carbon footprints, pension funds' funding ratios and dividends"]

4.3 The role of carbon footprint reporting by pension funds

In this section we analyze the effects of CO_2 reporting by pension funds. We use the fact that several pension funds have started reporting their CO_2 emissions recently, which enables us to test for recent periods whether the pension funds that report their CFPalso perform better in terms of CFP or whether this is just window-dressing (Gupta and Mason, 2016). In addition, by considering a subsample over the period 2013Q1-2015Q4 we analyze if our main results from Table 3 are sensitive to the time period of our sample as several studies suggest that the sample period may drive the results (e.g. Van Beurden and Gössling, 2008).¹⁸

Table 5 shows that the effect of $CarbonReporting_i$ is negative and significant, as expected. Pension funds that report CO_2 emissions related to their stock investments on average have a CFP that is 23 tonnes per million of sales lower. The effect is sizeable in economic terms and refutes the notion of window-dressing with regard to investors, although future work in this area is warranted as this analysis is based on only seven pension funds that report against 29 that do not report over a relatively short time period.

[Insert Table 5—"Role of carbon footprint reporting by pension funds(2013-2015)"]

As a robustness test, Table 5 indicates that in the recent subsample the size of the pension funds is still positively related to CFP. The size of the estimated coefficients is very similar to that of coefficients reported in Table 3. We also confirm that there is no relationship between total stock returns and CFP in the recent subsample. These results are in contrast with Ibikunle and Steffen (forthcoming) who compare the performance of green, black and conventional mutual funds and only find higher returns of green

¹⁸As this exercise significantly reduces the number of observations, the power of our statistical tests is much lower so we may expect that some relationships are rendered insignificant.

funds in their most recent subsample of 2012-2014. They argue that green funds' risk adjusted returns have progressively improved over time (see also Climent and Soriano, 2011; Ziegler et al., 2007), while we find no relationship between pension funds' *CFP* and total stock returns.

The relationship between dividend yields and CFPs remains significant for the period 2013-2015, although only in two of the presented specifications. However, the estimated size of the dividend effect is highly stable over time. The results for the market β are also significant in the more recent time period and suggest that pension fund portfolios with a higher β are associated with lower CFP. The results on active share are robust and comparable in size to what we found in Table 3. We find that the share of pension funds total portfolio allocated to stocks is positively related to CFP, although it is only significant at 10%. By contrast, Thistlethwaite (2015) finds a negative relationship between the percentage of assets allocated to equity and CFPs.

4.4 Controlling for industry weights

The CFP industry exposures in Table 1 show that there are significant differences in terms of the average CFP across different industries. This raises the question whether the variation in pension funds' exposures to particular industries is driving our results. Therefore, we include the contemporaneous portfolio industry weights (NACE level 1) as explanatory variables in Table 6. We take contemporaneous weights instead of lagged weights to prevent industry weights capturing portfolio rebalancing effects.

In Column (1) to (15) of Table 6 we add industry weights one by one to test whether a single industry is driving our results. We find that the inclusion of industry weights one by one leaves most of our main results unaffected, except Column (3) and Column (6). Table 6 shows that once we add the portfolio weight in 'Electricity, gas and steam' (industry D)—the industry with the highest CFP—several of our effects reduce in size or disappear. The effect size of *DividendYield* is much smaller and less significant for 'Electricity, gas and steam' and 'Wholesale and retail trade'. The portfolio weight in 'Electricity, gas and steam' appears to partially pick up the relationship between CFPand dividend yield, which is supported by the observation that industry D also has the highest dividend yield on average.¹⁹ Also in Column (3), β is now insignificant. The low β observed for 'Electricity, gas and steam' appears to support this result.²⁰ The effect of *ActiveShare* is much smaller and insignificant, which suggests that to some extent pension funds achieve a lower CFP by taking positions in 'Electricity, gas and steam' that are smaller than the ones in the benchmark.

[Insert Table 6—"Determinants of carbon footprints and industry portfolio weights"]

These results suggest that our main results on the trade-off between dividend yield and CFP and the trade-off between systematic risk and CFP are to some extent driven by the exposures to 'Electricity, gas and steam'. To further check this interpretation, we take a conservative specification that completely saturates the model with all industry weights, except the omitted category: 'Other' (Industry Z). Column (16) reports the results of this specification. The adjusted R-squared increases to 0.82, but with the exception of *Size* we can no longer confirm the results from our baseline specification in Table 3. Again, this appears to be mostly driven by 'Electricity, gas and steam, because once we omit the weight of 'Electricity, gas and steam' in Column (17) all our effects are again comparable in size and significance to the main effects reported in Table 3. These results confirm our initial finding that the identified trade-offs are to some extent driven by the exposure to 'Electricity, gas and steam'.²¹

This highlights the intricate link between CFP reduction and industry effects (Monasterolo et al., 2016; Koch and Bassen, 2013). CFP is concentrated in a few highly polluting industries which only comprise a small share of the total stock market value as well

 $^{^{19}{\}rm See}$ also Appendix Table C.1.

 $^{^{20}\}mathrm{See}$ also Appendix Table C.2.

 $^{^{21}}$ Note that once we rerun the the whole analysis without the stocks from the 'Electricity, gas and steam' industry that the main results do not change.

as pension fund's stock investments (see Appendix B). In line with this, Andersson et al. (2016) show that it is therefore possible to significantly reduce CFP, while maintaining minimum tracking error with the general market benchmark portfolio. Our results are consistent with their findings to the extent that trade-offs are largely driven by very few polluting industries.

5 Robustness checks

We perform two robustness checks for our dependent variable CFP. First, we check whether CFP outliers could be driving the main results in Table 3. Instead of CFP, we take the natural logarithm of CFP as dependent variable. Appendix Table E.1 shows that our results remain qualitatively similar to those reported in Table 3. Second, CFPis a measure of pension funds' absolute carbon footprint, but what might matter more is pension funds' CFP performance compared to a benchmark. Therefore, we reestimate the main analyzes in Table 3 and take as dependent variable the ratio of CFPto $CFP^{Benchmark}$ (see also Figure 1). Appendix Table E.2 shows that the sign and significance of the parameters is very similar to those reported in Table 3. Apparently, those variables that explain the variation of pension funds' CFP performance also account for the variation of $CFP/CFP^{Benchmark}$, even though they explain less variation as witnessed by Adjusted R^2 values that are noticeably lower than in Table 3.

6 Conclusion

In this paper we study a possible explanation for the variation of carbon footprints (CFP) across pension funds' equity portfolios. We combine Dutch pension funds' security-by-security equity holdings—obtained from supervisory data—with securities' carbon levels, returns, dividend yield and beta. We analyze this data at the portfolio level, supplementing it with the portfolio's active share, the share pension funds invest

in equity, their funding ratio, a variable indicating whether they report their CFP and industry weights.

Our main results indicate that pension funds face investment trade-offs when attempting to reduce their CFP. First, those that shift their portfolio towards high dividend yield stocks have lower CFP; a trade-off most pronounced for well-funded pension funds. Second, reduction in CFP is associated with pension funds deviating more from the benchmark and taking more systematic risk (β). These results suggest that lowering CFP could be contentious as it is associated with lower expected dividend returns, a higher active share and more systematic risk. To some extent these trade-offs seem to be driven by industry exposures to energy and utilities, which suggests that pension funds could significantly lower CFP without sacrificing risk-adjusted returns by divesting from a limited number of firms (Andersson et al., 2016).

This study is the first to calculate and analyze the CFP of pension funds' equity investment from disaggregated data. A critical element of this approach is the measurement and availability of CO_2 data. There is a wider call to increase the transparency (Liesen et al., 2015; Thistlethwaite, 2015) and quality (FSB, 2016; Gupta and Mason, 2016) of firm level CO_2 emissions. Firm information on CO_2 exposures allows investors to more accurately calculate their portfolio CFP. In this study, we find that those pension funds that calculate and disclose their CFP have a significantly lower CFP than those that refrain from doing so, suggesting measurement implies better management.

References

- Allen, F., Bernardo, A. E. and Welch, I. (2000). A theory of dividends based on tax clienteles, *The Journal of Finance* 55(6): 2499–2536.
- An, H., Huang, Z. and Zhang, T. (2013). What determines corporate pension fund risk-taking strategy?, Journal of Banking & Finance 37(2): 597–613.
- Andersson, M., Bolton, P. and Samama, F. (2016). Hedging climate risk, Financial Analysts Journal 72(3): 13–32.
- Andonov, A., Bauer, R. and Cremers, M. (2015). Pension fund asset allocation and liability discount rates, *Mimeo*.
- Bams, D., Schotman, P. C. and Tyagi, M. (2016). Pension fund asset allocation in low interest rate environment, *Mimeo*.
- Bauer, R., Koedijk, K. and Otten, R. (2005). International evidence on ethical mutual fund performance and investment style, *Journal of Banking & Finance* **29**(7): 1751–1767.
- Bohl, M. T., Kaufmann, P. and Siklos, P. L. (2015). What drove the mid-2000s explosiveness in alternative energy stock prices? Evidence from US, European and global indices, *International Review of Financial Analysis* 40: 194–206.
- Bradley, D., Pantzalis, C. and Yuan, X. (2016). The influence of political bias in state pension funds, *Journal of Financial Economics* 119(1): 69–91.
- Broeders, D., Chen, D. H., Minderhoud, P. and Schudel, W. (2016). Pension funds herding, DNB Working Paper No. 503.
- Climent, F. and Soriano, P. (2011). Green and good? The investment performance of US environmental mutual funds, *Journal of Business Ethics* 103(2): 275–287.

- Cortez, M. C., Silva, F. and Areal, N. (2009). The performance of European socially responsible funds, *Journal of Business Ethics* 87(4): 573–588.
- Cortez, M. C., Silva, F. and Areal, N. (2012). Socially responsible investing in the global market: The performance of US and European funds, *International Journal of Finance & Economics* 17(3): 254–271.
- Cremers, K. M. and Petajisto, A. (2009). How active is your fund manager? A new measure that predicts performance, *Review of Financial Studies* **22**(9): 3329–3365.
- Dahlquist, M., Robertsson, G. and Rydqvist, K. (2014). Direct evidence of dividend tax clienteles, *Journal of Empirical Finance* 28: 1–12.
- de Dreu, J. and Bikker, J. A. (2012). Investor sophistication and risk taking, Journal of Banking & Finance 36(7): 2145–2156.
- De Haan, L. (forthcoming). Recovery measures of underfunded pension funds: Higher contributions, no indexation or pension cuts?, *Journal of Pension Economics and Finance*.
- De Haan, M., Dam, L. and Scholtens, B. (2012). The drivers of the relationship between corporate environmental performance and stock market returns, *Journal of Sustainable Finance & Investment* 2(3-4): 338–375.
- De Jong, M. and Nguyen, A. (2016). Weathered for climate risk: A bond investment proposition, *Financial Analysts Journal* 72(3): 34–39.
- Derwall, J., Guenster, N., Bauer, R. and Koedijk, K. (2005). The eco-efficiency premium puzzle, *Financial Analysts Journal* 61(2): 51–63.
- Dimson, E., Karakaş, O. and Li, X. (2015). Active ownership, Review of Financial Studies 28(12): 3225–3268.

- Duijm, P. and Steins Bisschop, S. S. (2015). Short-termism of long-term investors? The investment behaviour of Dutch insurance companies and pension funds, DNB Working Paper No. 498.
- Dyck, I., Lins, K. V., Roth, L. and Wagner, H. F. (2016). Do institutional investors transplant social norms? International evidence on corporate social responsibility, *Mimeo*.
- Ferson, W. and Khang, K. (2002). Conditional performance measurement using portfolio weights: Evidence for pension funds, *Journal of Financial Economics* 65(2): 249–282.
- FSB (2016). Recommendations of the task force on climate-related financial disclosures (TCFD).
- Galema, R., Plantinga, A. and Scholtens, B. (2008). The stocks at stake: Return and risk in socially responsible investment, *Journal of Banking & Finance* **32**(12): 2646–2654.
- Griffin, P. A., Jaffe, A. M., Lont, D. H. and Dominguez-Faus, R. (2015). Science and the stock market: Investors' recognition of unburnable carbon, *Energy Economics* 52: 1–12.
- Gupta, A. and Mason, M. (2016). Disclosing or obscuring? The politics of transparency in global climate governance, *Current Opinion in Environmental Sustainability* 18: 82– 90.
- Hoepner, A. G. and Schopohl, L. (2016). On the price of morals in markets: An empirical study of the Swedish AP-funds and the Norwegian government pension fund, *Journal* of Business Ethics pp. 1–28.

Hong, H. G., Li, F. W. and Xu, J. (2016). Climate risks and market efficiency, *Mimeo*.Ibikunle, G. and Steffen, T. (forthcoming). European green mutual fund performance:

a comparative analysis with their conventional and black peers, *Journal of Business Ethics* pp. 1–19.

- Kempf, A. and Osthoff, P. (2008). SRI funds: Nomen est omen, Journal of Business Finance & Accounting 35(9-10): 1276–1294.
- Koch, N. and Bassen, A. (2013). Valuing the carbon exposure of European utilities: The role of fuel mix, permit allocation and replacement investments, *Energy Economics* 36: 431–443.
- Kreander, N., Gray, R. H., Power, D. M. and Sinclair, C. D. (2005). Evaluating the performance of ethical and non-ethical funds: A matched pair analysis, *Journal of Business Finance & Accounting* **32**(7-8): 1465–1493.
- Liesen, A., Hoepner, A. G., Patten, D. M. and Figge, F. (2015). Does stakeholder pressure influence corporate GHG emissions reporting? Empirical evidence from Europe, *Accounting, Auditing & Accountability Journal* 28(7): 1047–1074.
- Margolis, J. D. and Walsh, J. P. (2003). Misery loves companies: Rethinking social initiatives by business, Administrative Science Quarterly 48(2): 268–305.
- McCahery, J. A., Sautner, Z. and Starks, L. T. (forthcoming). Behind the scenes: The corporate governance preferences of institutional investors, *The Journal of Finance*.
- Mohan, N. and Zhang, T. (2014). An analysis of risk-taking behavior for public defined benefit pension plans, *Journal of Banking & Finance* 40: 403–419.
- Monasterolo, I., Battiston, S., Janetos, A. and Zheng, Z. (2016). Understanding investors' exposure to climate stranded assets to inform the post-carbon policy transition in the eurozone, *Mimeo*.
- Nofsinger, J. and Varma, A. (2014). Socially responsible funds and market crises, *Journal* of Banking & Finance 48: 180–193.

- Orlitzky, M., Schmidt, F. L. and Rynes, S. L. (2003). Corporate social and financial performance: A meta-analysis, Organization Studies 24(3): 403–441.
- PGGM (2016). De CO2 voetafdruk van de aandelenportefeuille.
- Rauh, J. D. (2009). Risk shifting versus risk management: Investment policy in corporate pension plans, *Review of Financial Studies* 22(7): 2687–2733.
- Renneboog, L., Ter Horst, J. and Zhang, C. (2008). Socially responsible investments: Institutional aspects, performance, and investor behavior, *Journal of Banking & Finance* **32**(9): 1723–1742.
- Rezec, M. and Scholtens, B. (2016). Financing energy transformation: The role of renewable energy equity indices, *International Journal of Green Energy* (forthcoming).
- Scholtens, B. (2006). Finance as a driver of corporate social responsibility, Journal of Business Ethics 68(1): 19–33.
- Shin, H., Ellinger, A. E., Nolan, H. H., DeCoster, T. D. and Lane, F. (forthcoming). An assessment of the association between renewable energy utilization and firm financial performance, *Journal of Business Ethics* pp. 1–18.
- Short, H., Zhang, H. and Keasey, K. (2002). The link between dividend policy and institutional ownership, *Journal of Corporate Finance* 8(2): 105–122.
- Sievanen, R., Rita, H. and Scholtens, B. (2013). The drivers of responsible investment: The case of European pension funds, *Journal of Business Ethics* 117(1): 137–151.
- Silva, F. and Cortez, M. C. (2016). The performance of US and European green funds in different market conditions, *Journal of Cleaner Production* 135: 558–566.
- Thistlethwaite, J. (2015). The politics of experimentation in climate change risk reporting: The emergence of the climate disclosure standards board (CDSB), *Environmental Politics* **24**(6): 970–990.

- Thomson Reuters (2016). Thomson Reuters ESG Carbon Data and Estimate Models. URL: http://financial.thomsonreuters.com/content/dam/openweb/documents/pdf/financial/esg-carbon-data-estimate-models.pdf, 2017-01-06
- Van Beurden, P. and Gössling, T. (2008). The worth of values: A literature review on the relation between corporate social and financial performance, *Journal of Business Ethics* 82(2): 407–424.
- Ziegler, A., Schröder, M. and Rennings, K. (2007). The effect of environmental and social performance on the stock performance of European corporations, *Environmental and Resource Economics* 37(4): 661–680.

Industry	Mean	Std.	Min	Max	Ν	n
B. Mining and quarrying	2,087.9	2,921.8	625.1	9,408.2	2,057	414
C. Manufacturing	275.3	63.3	197.1	407.5	8,031	1,389
D. Electricity, gas, steam	2,963.6	149.7	2,641.6	3,099.3	1,072	197
E. Water supply; sewerage, waste management	732.8	123.1	639.2	991.2	152	29
F. Construction	433.2	125.8	265.1	628.0	1,026	172
G. Wholesale and retail trade	62.0	9.5	47.5	80.0	1,613	294
H. Transportation and storage	615.5	64.8	498.1	713.0	971	169
I. Accommodation and food service activities	183.1	26.4	143.2	223.6	259	46
J. Information and communication	46.8	7.0	35.9	57.8	2,234	433
K. Financial and insurance activities	43.3	9.9	30.8	62.9	4,215	806
L. Real estate activities	467.5	484.4	137.7	$1,\!607.5$	564	195
M. Professional, scientific and technical activities	326.1	97.8	205.3	475.8	1,969	428
N. Administrative and support service activities	209.3	52.6	139.0	320.6	388	74
R. Arts, entertainment and recreation	68.5	17.8	55.1	108.1	243	44
Z. Other	329.2	242.4	97.7	774.2	1,218	247
Total	487.3	1,106.1	43.9	677.9	26,012	4,657

 Table 1: Summary statistics: security-level CFP by industry

This table reports summary statistics of security-level CFP for a panel dataset of all stocks in our sample (2009Q1-2015Q4), weighted by market value to total industry market value. Portfolio weights are rebalanced each quarter for each industry. Industries are classified according to NACE level 1. Industry Z. 'Other' is created and consists of the following industries with limited stock market presence: A. 'Agriculture, forestry and fishing', O. 'Public administration and defence; compulsory social security', P. 'Education', Q. 'Human health and social work activities' and S. 'Other service activities'. N indicates the number of observations, n indicates the number of stocks.

Variable	Mean	Std.	Min	Max
$\overline{CO_2}$ in million tonnes	1.63	4.88	0.07	1.99
CFP_{it}	288.29	76.22	201.61	390.62
Total number of stocks	1108.81	801.64	162.50	2197.50
Total assets in billions of euros	19.83	48.40	2.12	31.39
$\log Totalassets_{it}$	15.77	1.18	14.53	17.25
$Returns_{it}in\%$	15.60	16.96	-4.02	34.39
$DividendYield_{it}$ in %	2.94	0.53	2.38	3.58
β_{it}	0.99	0.08	0.91	1.07
$ActiveShare_{it}$	34.98	8.45	22.51	45.84
$ShareEquity_{it}$ in %	30.93	8.15	19.37	40.97
$FundRatio_{it}$	1.09	0.12	0.95	1.25
$FundRatio_{it} > 1.05$	0.62	0.49	0.00	1.00
$\Delta_{+}FundRatio_{it} > 1.05$	0.08	0.28	0.00	0.00
CarbonReporting	0.19	0.40	0.00	1.00
Industry weights				
B. Mining and quarrying	7.59	2.76	4.08	10.78
C. Manufacturing	33.58	5.24	28.16	38.72
D. Electricity, gas, steam	3.39	1.48	1.88	4.98
E. Water supply; sewerage, waste management	0.19	0.21	0.00	0.37
F. Construction	1.93	1.73	0.63	3.41
G. Wholesale and retail trade	5.31	2.14	3.25	7.82
H. Transportation and storage	2.20	1.09	1.28	3.50
I. Accommodation and food service activities	0.80	0.57	0.19	1.48
J. Information and communication	9.93	2.51	7.02	12.46
K. Financial and insurance activities	19.59	4.79	14.80	25.04
L. Real estate activities	2.98	4.87	0.18	7.22
M. Professional, scientific and technical activities	6.93	2.15	5.06	9.65
N. Administrative and support service activities	0.89	0.71	0.29	1.94
R. Arts, entertainment and recreation	0.64	0.39	0.25	1.07
Z. Other	4.05	5.48	1.52	6.48

 Table 2: Summary statistics

This table reports summary statistics for the estimation sample: n=890; 41 pension funds over 2009Q1-2015Q4 (unbalanced). Except for the *CFP* variables and industry weights, all variables are lagged one quarter. *CFP* is measured in terms of tonnes of CO_2 per millions euros of sales.

	(1)	(2)	(3)	(4)	(5)	(6)
$\log TotalAssets_{it}$	14.2608***	14.6706^{***}	14.1575^{***}	14.6311***	* 12.7900 ^{***}	13.3307^{***}
	[4.821]	[4.901]	[4.982]	[4.742]	[4.133]	[4.110]
$Returns_{it}$		-0.9571	0.8770	0.7945	0.5531	0.5299
		[1.227]	[1.056]	[0.991]	[0.968]	[0.951]
$DividendYield_{it}$			39.9173^{***}	32.1008^{**}	29.5052^{**}	30.6811^{**}
			[12.486]	[12.832]	[12.206]	[11.829]
β_{it}				-197.1518^{**}	-226.7534^{**} -	-220.9720^{**}
				[90.749]	[84.084]	[86.780]
$ActiveShare_{it}$					-1.5140^{***}	-1.3766^{***}
					[0.389]	[0.400]
$ShareEquity_{it}$						-0.6489
						[0.635]
Constant	176.0790^{**}	145.3616^{*}	31.3381	239.0102^{*}	355.3267^{***}	349.6864^{***}
	[78.383]	[85.400]	[99.920]	[133.545]	[123.476]	[120.711]
Observations	890	890	890	890	890	890
Adjusted R^2	0.455	0.457	0.487	0.531	0.557	0.561
Number of pension funds	41	41	41	41	41	41

 Table 3: Main results: Determinants of carbon footprints

This table reports main effects of estimating equation 2. All explanatory variables are lagged by one period. All estimations include year-quarter fixed effects. Standard errors are clustered at the pension fund level and are reported in parentheses and ***, **, * correspond to the 1%, 5%, and 10% level of significance, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
$\log TotalAssets_{it}$	13.5975^{***}	13.7339^{***}	13.8925^{***}	13.7100^{***}	13.3272^{***}	[*] 13.4796 ^{***}
-	[4.093]	[4.059]	[4.166]	[4.047]	[4.116]	[4.056]
$Returns_{it}$	0.4372	0.4208	0.4733	0.4472	0.5450	0.6075
	[0.922]	[0.920]	[0.926]	[0.934]	[0.952]	[0.964]
$DividendYield_{it}$	31.8019**	-49.7358	32.8839^{**}	18.0289	30.6683**	26.6335^{**}
	[12.167]	[54.677]	[12.293]	[13.889]	[11.839]	[11.645]
β_{it}	-216.1469^{**}	-218.2207^{**}	-213.3187** -	-215.4248**	-221.4931**	-220.5942^{**}
	[88.015]	[88.227]	[87.760]	[87.513]	[86.778]	[86.766]
$ActiveShare_{it}$	-1.3317^{***}	-1.3412^{***}	-1.3388^{***}	-1.3062^{***}	-1.3756^{***}	-1.3787^{***}
	[0.389]	[0.394]	[0.388]	[0.387]	[0.400]	[0.396]
$ShareEquity_{it}$	-0.7452	-0.7002	-0.7037	-0.6655	-0.6525	-0.6614
1 0	[0.659]	[0.632]	[0.625]	[0.596]	[0.636]	[0.635]
$FundRatio_{it}$	43.6957	-174.5525				
	[37.806]	[140.969]				
$FundRatio_{it} \times$		75.2839				
$DividendYield_{it}$		[49.593]				
$FundRatio_{it} > 1.05$			15.1605	-55.0420		
			[10.078]	[33.552]		
$FundRatio_{it} > 1.05 \times$				23.6171^{**}		
$DividendYield_{it}$				[11.126]		
$\Delta_{+}FundRatio_{it} > 1.05$					4.0882	-135.2408^{***}
					[4.679]	[47.080]
$\Delta_{+}FundRatio_{it} > 1.05 \times$						46.4594^{***}
$DividendYield_{it}$						[16.166]
Constant	292.5806^{**}	538.5989^{**}	320.3421^{**}	379.0251^{***}	350.7378^{***}	366.5849 ^{***}
	[143.740]	[214.715]	[127.579]	[131.024]	[120.690]	[117.970]
Observations	890	890	890	890	890	890
Adjusted R^2	0.564	0.567	0.568	0.573	0.560	0.566
Number of pension funds	41	41	41	41	41	41

Table 4: Carbon footprints, pension funds' funding ratios and dividends

This table reports interactions of pension funds' funding ratio indicators with dividend yield. $FundRatio_{it}$ indicates the actual funding ratio. $FundRatio_{it} > 1.00$ and $\Delta_+FundRatio_{it} > 1.05$ are dummy variables indicating whether funding ratio is larger than 105 percent and whether a pension fund reaches a funding ratio larger than 105 percent, respectively. All explanatory variables are lagged by one period. All estimations include year-quarter fixed effects. Standard errors are clustered at the pension fund level and are reported in parentheses and ***, **, * correspond to the 1%, 5%, and 10% level of significance, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
$\log TotalAssets_{it}$	14.3409**	^{**} 14.3435 ^{**}	14.8679^{**}	** 12.1144	** 10.0549 [*]	** 8.2849*	** 13.1315 ^{***}
	[4.040]	[4.059]	[4.253]	[3.301]	[2.621]	[2.665]	[3.375]
$Returns_{it}$		-0.9661	0.9346	0.9434	0.2554	0.3355	0.1825
		[1.381]	[1.245]	[1.111]	[1.092]	[0.995]	[1.019]
$DividendYield_{it}$			42.3110^{**}	* 30.0382 $*$	20.5472	16.4431	19.9430
			[16.062]	[16.730]	[17.153]	[16.784]	[16.293]
β_{it}			-	-244.1726^{*}	-298.9879^{*}	-318.2261^{*}	-284.0480^{**}
				[131.077]	[122.819]	[111.826]	[108.770]
$ActiveShare_{it}$					-1.5856^{*}	$^{**}-1.9205^{*}$	$^{**}-1.9920^{***}$
					[0.421]	[0.490]	[0.511]
$ShareEquity_{it}$						1.2804^*	1.1188^*
						[0.661]	[0.642]
$CarbonReporting_{it}$							-23.0626^{**}
							[9.666]
Observations	397	397	397	397	397	397	397
Adjusted R^2	0.319	0.322	0.389	0.466	0.545	0.570	0.586
Number of pension funds	36	36	36	36	36	36	36

Table 5: Role of carbon footprint reporting by pension funds (period 2013-2015)

This table reports the main results of Table 4 for the subsample 2013-2015. In column (7) we add an additional explanatory variable CarbonReporting, which is an indicator variable equal to one if a pension fund reports its CFP and zero otherwise. All explanatory variables are lagged by one period. All estimations include year-quarter fixed effects. Standard errors are clustered at the pension fund level and are reported in parentheses and ***, **, * correspond to the 1%, 5%, and 10% level of significance, respectively.

Table 6: Determinants of carbon footprints and industry portfolio weights

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)
$\log TotalAssets_{it}$	13.18^{***} [3.71]	12.53^{***}	10.49^{***} [3.29]	12.74^{***}	15.15^{***} [4.53]	12.28^{***}	12.66^{***} [3.52]	13.56^{***}	14.63^{***} [3.78]	13.06^{***}	14.22^{***}	12.94^{***} [4.33]	13.14^{***} [4.22]	13.02^{***}	13.30^{***} [4.09]	7.66^{***}	10.54^{**}
$Returns_{it}$	0.94	0.24	0.70	0.38	1.00	0.20	0.97	0.50	0.88	0.49	1.54	0.67	0.45	0.73	0.52	0.35	0.80
$DividendYield_{it}$	25.63^{**} [11.94]	27.77^{**} [11.81]	16.77^{*} [9.01]	32.71^{***}	32.52^{***} [11.87]	18.23 [16.35]	51.44^{***}	$\begin{bmatrix} 0.00 \\ 35.86 \end{bmatrix}$	33.34 ^{***}	$^{\circ} 29.34^{**}$ $[12.87]$	46.35^{***} [11.74]	34.09^{**} [12.86]	23.93^{**} [11.75]	35.74^{***} [12.55]	30.78^{**} [11.83]	9.26 [11.30]	31.38^{**} [13.52]
β_{it}	-237.63^{***} [85.75]	-228.40^{***} [84.29]	45.31 [63.98]	-216.98^{**} [85.36]	-190.19^{**} [91.44]	-263.12^{***} [88.08]	-197.38^{**} [77.35]	-195.02^{*} [98.43]	-201.37^{**} [82.03]	-230.82^{**} [92.66]	-211.55^{**} [86.83]	-223.03^{**} [84.37]	-230.58^{***} [83.74]	-213.73^{**}	-221.32^{**} [86.64]	-29.26 [42.92]	-246.51^{***} [69.76]
$ActiveShare_{it}$	-1.11***	-1.43^{***}	-0.52	-1.35^{***}	-1.21^{***}	-1.22^{***}	-1.92^{***}	-1.35^{***}	-1.17^{**}	-1.36^{***}	-1.15^{***}	-1.27***	-1.15^{**}	-1.37^{***}	-1.38***	-0.33	-0.81^{*}
$Share Equity_{it}$	[0.39] -0.38 [0.58]	[0.43] -0.63 [0.63]	[0.33] -0.12 [0.47]	[0.39] -0.61 [0.63]	[0.44] -0.63 [0.61]	[0.39] -0.64 [0.60]	[0.52] -0.47 [0.62]	[0.41] -0.64 [0.64]	[0.44] -0.77 [0.59]	[0.38] -0.62 [0.64]	[0.41] -0.84 [0.59]	[0.42] -0.60 [0.67]	[0.44] -0.52 [0.70]	[0.41] -0.64 [0.63]	[0.40] -0.66 [0.65]	$\begin{bmatrix} 0.36 \end{bmatrix} \\ 0.04 \\ \begin{bmatrix} 0.30 \end{bmatrix}$	[0.47] -0.32 [0.48]
Industry B	6.75^{***} [2.48]	* *	[0.11]	[0.00]	[0.01]	[0.00]	[0:02]	[0.01]	[0.00]	[0.01]	[0.00]	[0.01]	[0.10]	[0.00]	[0.00]	3.05^{**} [1.48]	3.35^{*} [1.87]
Industry C	[]	-0.74 [1.20]														-1.90^{**} [0.73]	-2.73^{***} [0.91]
Industry D		[0]	27.74^{***} [3.40]													25.26^{***} [2.66]	[0:0-]
Industry E			. ,	31.59 [21.42]												41.58^{***} [14.85]	22.08 $[17.85]$
Industry F					-5.40^{**} [2.27]											-0.77 [1.48]	-2.82 [1.80]
Industry G					[·]	-4.66 [4.35]										-6.79^{**}	-7.22^{**} [3.00]
Industry H						[1:00]	15.54 ^{**}									8.71 ^{**}	13.69 ^{***}
Industry I							[0.10]	9.12 [10.19]								[3.30] 3.37 [6.20]	[4.36] 10.66 [7, 27]
Industry J								[10.13]	4.13							$\begin{bmatrix} 0.20 \end{bmatrix}$ 0.37 [1.21]	2.96^{*}
Industry K									[2.00]	0.38 $[1,22]$						-0.14 [0.70]	0.19
Industry L										[1.22]	-3.39^{***} [0.58]					-2.86^{***}	-3.70^{**}
Industry M											[0.00]	-2.33 [3.06]				2.54 [2.06]	0.96 [2.55]
Industry N												[0.00]	-9.31			-7.53	-14.15^{*} [7.46]
Industry R													[1.12]	16.36 [14.75]		-7.07 [6.18]	-0.62 [9.75]
Industry Z														[1110]	0.10 [0.47]	[0.10]	[0.10]
Constant	323.62^{***} [118.28]	400.05^{***} [136.52]	59.36 [97.82]	334.66^{***} [120.94]	302.85^{**} [127.73]	$\begin{array}{c} 464.21^{***} \\ [151.42] \end{array}$	250.29^{**} [123.24]	293.05^{*} [148.68]	258.94^{**} [124.11]	359.34^{***} [122.78]	$291.74^{**} \\ [116.45]$	356.35^{***} [116.45]	381.93^{***} [123.12]	323.89^{**} [129.92]	349.55^{***} [120.82]	235.16^{**} [97.78]	434.23 [146.52]
Observations Adjusted R^2 Number of pension fund	890 0.597 ds 41	890 0.562 41	890 0.738 41	890 0.567 41	890 0.573 41	890 0.572 41	890 0.593 41	890 0.564 41	890 0.578 41	890 0.561 41	890 0.597 41	890 0.563 41	890 0.566 41	890 0.566 41	890 0.560 41	890 0.820 41	890 0.704 41

This table reports our main estimates including contemporaneous industry weights as well. For an overview of the industry names, refer to Table 1. All explanatory variables are lagged by one period. All estimations include year-quarter fixed effects. Standard errors are clustered at the pension fund level and are reported in parentheses and ***, **, * correspond to the 1%, 5%, and 10% level of significance, respectively.



Figure 1: Average pension fund footprint versus a market benchmark

Appendix A Sample coverage of CO_2 data

Size quantiles	2009	2010	2011	2012	2013	2014	2015	Total
1	71.4	79.9	81.7	80.4	83.6	85.2	87.0	83.1
2	73.6	80.6	81.8	81.1	83.8	86.5	87.8	83.5
3	73.5	80.5	82.1	82.3	84.5	87.0	88.2	83.8
4	71.7	79.3	81.2	82.3	84.5	86.8	88.3	83.3
5	70.4	77.8	80.0	81.6	83.5	86.3	88.3	82.3
6	69.8	77.6	79.3	81.3	82.5	85.9	88.2	81.6
7	69.6	76.7	79.1	81.0	82.3	86.1	88.5	81.3
8	68.5	76.6	79.0	81.1	82.9	86.4	88.7	81.6
9	67.4	76.5	79.3	81.1	82.8	86.3	88.7	81.8
10	63.3	75.0	77.9	80.9	82.5	86.3	88.8	81.6
Total	69.8	77.8	80.1	81.3	83.3	86.3	88.2	82.4

Table A.1: Percentage	of equity	investments	covered	by	CO_2	data
-----------------------	-----------	-------------	---------	----	--------	------

This table reports the percentage of equity investment coverage by CO_2 data according to pension fund size quantiles and years. Size quantiles are based on pension funds' total assets.

Appendix B Summary statistics security-level data

Industry	Moon	Std	n10	n00	N
	Wiean	Stu.	p10	p30	19
B. Mining and quarrying	2,548.0	313.3	1,989.5	2,957.9	2,057
C. Manufacturing	10,793.5	$2,\!470.9$	7,229.6	$14,\!622.5$	8,031
D. Electricity, gas, steam	1,275.1	54.1	$1,\!193.4$	1,352.9	1,072
E. Water supply; sewerage, waste management	75.6	18.8	50.3	107.9	152
F. Construction	580.9	134.8	413.9	794.3	1,026
G. Wholesale and retail trade	1,757.4	567.8	954.1	2,649.2	$1,\!613$
H. Transportation and storage	820.8	218.0	539.2	1,122.3	971
I. Accommodation and food service activities	267.9	77.4	137.4	368.9	259
J. Information and communication	$3,\!403.4$	966.3	2,281.6	5,077.0	2,234
K. Financial and insurance activities	5,772.7	1,326.3	4,062.1	7,509.0	4,215
L. Real estate activities	470.8	379.2	164.7	1,024.4	564
M. Professional, scientific and technical activities	2,185.7	928.9	1,202.1	$3,\!890.5$	1,969
N. Administrative and support service activities	257.3	86.3	124.9	349.0	388
R. Arts, entertainment and recreation	244.5	74.2	139.9	387.5	243
Z. Other (A, O, P, Q, S)	737.9	212.4	419.3	1,077.7	1,218

Table B.1:	Summary	statistics	security-level	market	values	by industry
Table D.I.	Summary	5040150105	security level	markee	varues	by maasury

This table reports summary statistics for stocks' market values in billions of euros, summarized over 2009Q1-2015Q4. Industries are classified according to NACE level 1. Industry Z. 'Other' is created and consists of the following industries with limited coverage: A. 'Agriculture, forestry and fishing', O. 'Public administration and defence; compulsory social security', P. 'Education', Q. 'Human health and social work activities' and S. 'Other service activities'.

Table B.2: Summary statistics security-level pension fund stock holdings by industry

Industry	Mean	Std.	p10	p90	Ν
B. Mining and quarrying	14.4	1.5	11.7	16.3	2,057
C. Manufacturing	67.2	20.2	39.2	94.0	8,031
D. Electricity, gas, steam	6.9	1.8	5.0	9.8	1,072
E. Water supply; sewerage, waste management	0.4	0.2	0.2	0.7	152
F. Construction	5.4	1.0	3.9	6.5	1,026
G. Wholesale and retail trade	9.5	3.6	4.3	14.9	$1,\!613$
H. Transportation and storage	4.3	1.7	2.2	6.9	971
I. Accommodation and food service activities	1.3	0.6	0.5	2.1	259
J. Information and communication	19.2	7.2	11.1	30.8	2,234
K. Financial and insurance activities	44.3	11.3	28.7	63.4	4,215
L. Real estate activities	15.1	12.5	5.4	33.4	564
M. Professional, scientific and technical activities	14.8	6.5	7.6	26.0	1,969
N. Administrative and support service activities	1.7	0.5	1.0	2.2	388
R. Arts, entertainment and recreation	1.3	0.5	0.6	2.1	243
Z. Other (A, O, P, Q, S)	4.7	1.8	2.3	7.8	$1,\!218$

This table reports security-level summary statistics for pension funds' total stock holdings in billions of euros, summarized over 2009Q1-2015Q4. Industries are classified according to NACE level 1. Industry Z. Other includes: A. Agriculture, forestry and fishing, O. Public administration and defence; compulsory social security, P. Education, Q. Human health and social work activities and S. Other service activities.

Industry	Mean	Std.	p10	p90	Ν
B. Mining and quarrying	2,835.4	430.8	1,944.0	3,286.6	2,057
C. Manufacturing	9,213.1	1,043.8	7,027.6	10,490.2	8,031
D. Electricity, gas, steam	1,243.1	107.1	1,056.7	$1,\!401.5$	1,072
E. Water supply; sewerage, waste management	58.0	9.4	43.9	67.1	152
F. Construction	674.2	106.1	520.7	850.8	1,026
G. Wholesale and retail trade	2,939.6	544.3	2,078.7	$3,\!817.0$	$1,\!613$
H. Transportation and storage	857.2	113.7	649.9	1,025.2	971
I. Accommodation and food service activities	120.8	22.9	85.9	156.4	259
J. Information and communication	1,993.5	246.0	$1,\!625.3$	$2,\!475.7$	2,234
K. Financial and insurance activities	4,642.4	568.6	3,563.2	5,338.0	4,215
L. Real estate activities	102.4	62.9	46.4	191.4	564
M. Professional, scientific and technical activities	$2,\!172.0$	504.8	$1,\!432.5$	3,015.9	1,969
N. Administrative and support service activities	192.3	19.6	150.8	215.1	388
R. Arts, entertainment and recreation	111.2	17.0	90.9	145.3	243
Z. Other (A, O, P, Q, S)	783.4	140.5	532.0	997.8	1,218

Table B.3: Summary statistics security level sales by industry

This table reports summary statistics for stocks' sales in millions of euros, summarized over 2009Q1-2015Q4. Industries are classified according to NACE level 1. Industry Z. 'Other' is created and consists of the following industries with limited coverage: A. 'Agriculture, forestry and fishing', O. 'Public administration and defence; compulsory social security', P. 'Education', Q. 'Human health and social work activities' and S. 'Other service activities'.

Industry	Mean	Std.	p10	p90	Ν
B. Mining and quarrying	1,418.0	101.0	1,285.9	1,607.5	2,057
C. Manufacturing	2,750.1	275.1	2,147.3	3,056.1	8,031
D. Electricity, gas, steam	3,277.9	262.5	2,814.3	$3,\!547.7$	1,072
E. Water supply; sewerage, waste management	36.6	1.7	33.6	39.1	152
F. Construction	228.0	53.7	116.3	285.4	1,026
G. Wholesale and retail trade	177.3	40.6	121.2	229.5	$1,\!613$
H. Transportation and storage	574.8	51.7	470.4	634.8	971
I. Accommodation and food service activities	18.6	2.3	14.0	21.3	259
J. Information and communication	95.8	5.8	88.7	106.4	2,234
K. Financial and insurance activities	158.4	32.5	129.7	222.3	4,215
L. Real estate activities	35.7	32.9	6.3	75.7	564
M. Professional, scientific and technical activities	764.4	95.4	584.6	896.3	1,969
N. Administrative and support service activities	40.0	8.1	31.7	51.6	388
R. Arts, entertainment and recreation	7.4	0.7	6.5	8.6	243
Z. Other (A, O, P, Q, S)	106.0	62.1	43.7	206.5	1,218

Table B.4: Summary statistics security-level CO_2 by industry

This table reports summary statistics for stocks' absolute CO_2 emissions in millions of tonnes, summarized over 2009Q1-2015Q4. Industries are classified according to NACE level 1. Industry Z. Other includes: A. Agriculture, forestry and fishing, O. Public administration and defence; compulsory social security, P. Education, Q. Human health and social work activities and S. Other service activities.

Appendix C Dividend yield and β by industry

Industry	Mean	Std.	p10	p90	Ν
B. Mining and quarrying	3.0	0.7	1.9	4.4	2,057
C. Manufacturing	2.3	0.2	2.1	2.6	8,031
D. Electricity, gas, steam	3.7	0.3	3.2	4.3	1,072
E. Water supply; sewerage, waste management	3.5	0.4	2.8	3.9	152
F. Construction	2.3	0.3	2.0	2.8	1,026
G. Wholesale and retail trade	1.8	0.1	1.7	2.1	1,613
H. Transportation and storage	2.1	0.3	1.8	2.8	971
I. Accommodation and food service activities	2.2	0.2	1.8	2.5	259
J. Information and communication	2.9	0.4	2.4	3.5	2,234
K. Financial and insurance activities	2.9	0.3	2.4	3.6	4,215
L. Real estate activities	3.7	0.3	3.4	4.2	564
M. Professional, scientific and technical activities	2.3	0.3	2.1	2.8	1,969
N. Administrative and support service activities	1.3	0.2	1.0	1.6	388
R. Arts, entertainment and recreation	2.1	0.5	1.5	2.8	243
Z. Other (A, O, P, Q, S)	1.9	0.2	1.5	2.3	1,218

Table C.1:	Summary	statistics	security-level	dividend	yield	by	industry
------------	---------	------------	----------------	----------	-------	----	----------

This table reports summary statistics of security-level *DividendYield* for a panel dataset of all stocks in our sample (2009Q1-2015Q4), weighted by market value to total industry market value. Portfolio weights are rebalanced each quarter, for each industry. Industries are classified according to NACE level 1. Industry Z. 'Other' is created and consists of the following industries with limited coverage: A. 'Agriculture, forestry and fishing', O. 'Public administration and defence; compulsory social security', P. 'Education', Q. 'Human health and social work activities' and S. 'Other service activities'.

Table C.2: Summary statistics security-level β by industry

Industry	Mean	Std.	p10	p90	Ν
B. Mining and quarrying	1.2	0.0	1.1	1.2	2,057
C. Manufacturing	1.0	0.0	0.9	1.0	8,031
D. Electricity, gas, steam	0.7	0.0	0.7	0.8	1,072
E. Water supply; sewerage, waste management	0.6	0.0	0.5	0.6	152
F. Construction	1.3	0.1	1.1	1.3	1,026
G. Wholesale and retail trade	0.9	0.0	0.8	0.9	$1,\!613$
H. Transportation and storage	0.9	0.0	0.8	1.0	971
I. Accommodation and food service activities	0.9	0.1	0.8	1.0	259
J. Information and communication	0.8	0.0	0.8	0.9	2,234
K. Financial and insurance activities	1.3	0.1	1.2	1.4	4,215
L. Real estate activities	1.0	0.1	0.8	1.2	564
M. Professional, scientific and technical activities	0.9	0.0	0.9	1.0	1,969
N. Administrative and support service activities	1.1	0.1	1.1	1.3	388
R. Arts, entertainment and recreation	1.2	0.1	1.0	1.4	243
Z. Other (A, O, P, Q, S)	1.0	0.1	0.9	1.1	1,218

This table reports summary statistics of security-level β for a panel dataset of all stocks in our sample (2009Q1-2015Q4), weighted by market value to total industry market value. Portfolio weights are rebalanced each quarter, for each industry. Industries are classified according to NACE level 1. Industry Z. 'Other' is created and consists of the following industries with limited coverage: A. 'Agriculture, forestry and fishing', O. 'Public administration and defence; compulsory social security', P. 'Education', Q. 'Human health and social work activities' and S. 'Other service activities'.

	1	2	3	4	5	6
1. $\log TotalAssets_{it}$	1					
2. $Returns_{it}$	0.0923^{***}	1				
3. $DividendYield_{it}$	-0.1043	-0.5967^{\ast}	1			
4. β_{it}	0.0335	0.1225^{**}	-0.2286^{***}	1		
5. $ActiveShare_{it}$	-0.2051^{***}	-0.0571	0.048	-0.1736^{***}	1	
6. $ShareEquity_{it}$	0.0547	0.0686^{**}	0.0049	0.0685^{**}	0.1954^{***}	1

Table D.1:	Correlations
------------	--------------

This table reports pairwise correlations between our main independent variables: ***, **, * correspond to the 1%, 5%, and 10% level of significance, respectively (n=890).

Appendix E Robustness checks

	(1)	(2)	(3)	(4)	(5)	(6)
$\log TotalAssets_{it}$	0.0557***	0.0567***	0.0551***	0.0564***	0.0477***	0.0489***
$Returns_{it}$	[0.018]	[0.018] -0.0023	$\begin{bmatrix} 0.018 \end{bmatrix} \\ 0.0034 \end{bmatrix}$	$\begin{bmatrix} 0.017 \end{bmatrix} \\ 0.0032 \end{bmatrix}$	$\begin{bmatrix} 0.014 \end{bmatrix} \\ 0.0020$	$\begin{bmatrix} 0.015 \end{bmatrix} \\ 0.0020 \end{bmatrix}$
$DividendYield_{it}$		[0.005]	$egin{array}{c} [0.004] \ 0.1231^{**} \end{array}$	$egin{array}{c} [0.004] \ 0.1016^{**} \end{array}$	$[0.004] \\ 0.0894^*$	$\frac{[0.004]}{0.0920}^{**}$
β_{it}			[0.049]	$egin{array}{c} [0.049] \\ -0.5421^{*} \end{array}$	$[0.047] \\ -0.6816^{**}$	$[0.046] \\ -0.6689^{**}$
$ActiveShare_{it}$				[0.310]	$[0.273] \\ -0.0071^{***}$	$[0.279] -0.0068^{***}$
$ShareEquity_{it}$					[0.002]	[0.002] -0.0014
Constant	5.1041^{***} [0.284]	5.0312^{***} [0.302]	4.6796^{***} [0.348]	5.2507^{***} [0.471]	5.7986^{***} [0.414]	$[0.002] \\ 5.7862^{***} \\ [0.406]$
Observations	890	890	890	890	890	890
Adjusted R^2	0.438	0.438	0.461	0.487	0.532	0.534
Number of pension funds	41	41	41	41	41	41

Table E.1: Robustness check: log CFP instead of CFP as dependent variable

This table reports main effects of a robustness check in which we replace the dependent variable in equation 2, CFP_{it} with the natural logarithm of this variable, $\log CFP_{it}$. All explanatory variables are lagged by one period. All estimations include year-quarter fixed effects. Standard errors are clustered at the pension fund level and are reported in parentheses and ***, **, * correspond to the 1%, 5%, and 10% level of significance, respectively.

Table E.2: Robustness check: $CFP_{it}/CFP_{it}^{Benchmark}$ as dependent variable

	(1)	(2)	(3)	(4)	(5)	(6)
$\log TotalAssets_{it}$	0.0429^{***}	0.0439^{***}	0.0424^{***}	0.0439^{***}	0.0380^{***}	0.0391^{***}
	[0.014]	[0.014]	[0.014]	[0.014]	[0.012]	[0.012]
$Returns_{it}$		-0.0024	0.0029	0.0026	0.0019	0.0018
		[0.004]	[0.003]	[0.003]	[0.003]	[0.003]
$DividendYield_{it}$			0.1159^{***}	0.0915^{**}	0.0832^{**}	0.0855^{**}
			[0.039]	[0.040]	[0.038]	[0.037]
β_{it}				-0.6142^{**}	-0.7088^{***}	-0.6977^{***}
				[0.256]	[0.235]	[0.242]
$ActiveShare_{it}$					-0.0048^{***}	-0.0046^{***}
					[0.001]	[0.001]
$ShareEquity_{it}$						-0.0013
					**	[0.002]
Constant	0.2439	0.1657	-0.1653	0.4817	0.8534^{**}	0.8425^{**}
	[0.221]	[0.248]	[0.289]	[0.361]	[0.335]	[0.329]
Observations	890	890	890	890	890	890
Adjusted R^2	0.087	0.090	0.143	0.229	0.283	0.286
Number of pension funds	41	41	41	41	41	41

This table reports main effects of a robustness check in which we replace the dependent variable in equation 2, CFP_{it} with a benchmark-adjusted dependent variable: $CFP_{it}/CFP_{it}^{Benchmark}$. All explanatory variables are lagged by one period. All estimations include year-quarter fixed effects. Standard errors are clustered at the pension fund level and are reported in parentheses and ***, **, * correspond to the 1%, 5%, and 10% level of significance, respectively.

Previous DNB Working Papers in 2017

- No. 542 Jasper de Jong, Marien Ferdinandusse and Josip Funda, Public capital in the 21st century: As productive as ever?
- No. 543 Martijn Boermans and Sweder van Wijnbergen, Contingent convertible bonds: Who invests in European CoCos?
- No. 544 Yakov Ben-Haim, Maria Demertzis and Jan Willem Van den End, Fundamental uncertainty and unconventional monetary policy: an info-gap approach
- No. 545 **Thorsten Beck and Steven Poelhekke**, Follow the money: Does the financial sector intermediate natural resource windfalls?
- No. 546 Lola Hernandez, Robbert-Jan 't Hoen and Juanita Raat, Survey shortcuts? Evidence from a payment diary survey
- No. 547 **Gosse Alserda, Jaap Bikker and Fieke van der Lecq**, X-efficiency and economies of scale in pension fund administration and investment
- No. 548 **Ryan van Lamoen, Simona Mattheussens, and Martijn Dröes**, Quantitative easing and exuberance in government bond markets: Evidence from the ECB's expanded asset purchase program
- No. 549 **David-Jan Jansen and Matthias Neuenkirch**, News consumption, political preferences, and accurate views on inflation
- No. 550 Maaike Diepstraten and Carin van der Cruijsen, To stay or go? Consumer bank switching behaviour after government interventions
- No. 551 Dimitris Christelis, Dimitris Georgarakos, Tullio Jappelli, Luigi Pistaferri and Maarten van Rooij, Asymmetric consumption effects of transitory income shocks
- No. 552 **Dirk Gerritsen, Jacob Bikker and Mike Brandsen**, Bank switching and deposit rates: Evidence for crisis and non-crisis years
- No. 553 Svetlana Borovkova, Evgeny Garmaev, Philip Lammers and Jordi Rustige, SenSR: A sentiment-based systemic risk indicator

DeNederlandscheBank

De Nederlandsche Bank N.V. Postbus 98, 1000 AB Amsterdam 020 524 91 11 dnb.nl